Author’s response to reviews

Title: Major vault protein suppresses lung cancer cell proliferation by inhibiting STAT3 signaling pathway

Authors:
Hui Bai (huibai@njmu.edu.cn)
Chenchen Wang (wcc0113@sina.com)
Lili Chen (514468494@qq.com)
Bin Jiang (jiangbin@njmu.edu.cn)
Nan Li (2397648344@qq.com)
Jin Xu (jinxu@njmu.edu.cn)
Xudong Zhu (zhuxudong@njmu.edu.cn)
Hanwen Zhang (hanwenzhang@njmu.edu.cn)
Xiaoyu Li (xyli@njmu.edu.cn)
Qing Yang (qyang@njmu.edu.cn)
Junqing Ma (1289366375@qq.com)
Yong Xu (yxu@njmu.edu.cn)
Jingjing Ben (bjj@njmu.edu.cn)
Qi Chen (qichen@njmu.edu.cn)

Version: 1 Date: 15 May 2018

Author’s response to reviews:

Dear Editor,

Enclosed please find our manuscript entitled “Major vault protein suppresses lung cancer cell proliferation by inhibiting STAT3 signaling pathway”, which was submitted to BMC Cancer for publication. It has neither been published nor is currently under consideration for publication by any
other journal. All the co-authors have read the manuscript and approved its submission to BMC Cancer.

The recommended reviewers are as follows:

Liuqing Yang  
Department of Molecular and Cellular Oncology, The University of Texas MD Anderson Cancer Center, Houston, Texas, USA.  
E-mail: lyang7@mdanderson.org

Keping Xie  
Department of Gastroenterology, Hepatology and Nutrition, The University of Texas MD Anderson Cancer Center, Houston, Texas, USA.  
E-mail: kepxie@mdanderson.org

Sincerely yours,

Dr. Qi Chen  
Professor of pathophysiology  
Department of Pathophysiology, Key Laboratory of Cardiovascular Disease and Molecular Intervention, Nanjing Medical University, Nanjing 210029, P. R. China  
Tel: 86 25 8686 2610  
Fax: 86 25 8650 8960  
Email: qichen@njmu.edu.cn

Editor's comments:  
1 - In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/about/editorialpolicies#StandardsofReporting), could you please ensure your manuscript reporting adheres to the ARRIVE guidelines (http://www.nc3rs.org.uk/arrive-guidelines) for the reporting of animal experiments. This is so your methodology can be fully evaluated and utilised. 
Could you please include a completed ARRIVE checklist as an additional file when submitting your revised manuscript. 
A: Yes, we have completed the ARRIVE checklist.

2 - Please clarify in the "Ethics approval and consent to participate" section whether the consent obtained was informed and whether written or verbal. If verbal, please indicate the reason for obtaining verbal rather than written consent and whether the ethics committee approved this form of consent. 
A: We have clarified the informed written consents were from the patients or their guardians in the "Ethics approval and consent to participate" section.

3 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section. 
A: Yes, we added “No funding body had any role in the design of the study, in the collection, analysis
and interpretation of data or in writing the manuscript” in the Funding section.

4 - Please check our authorship criteria and ensure all named authors have provided sufficient input to be so registered. http://www.biomedcentral.com/submissions/editorial-policies#authorship
A: Yes, we have confirmed the authorship.